Author's response to reviews

Title: Did HealthKick, a randomised controlled trial primary school nutrition intervention improve dietary quality of children in low-income settings in South Africa?

Authors:

Nelia P Steyn (nelia.steyn@uct.ac.za)
Anniza de Villiers (anniza.devilliers@mrc.ac.za)
Nomonde E Gwebushe (nomonde.gwebushe@mrc.ac.za)
Catherine Draper (catherine.draper@uct.ac.za)
Jillian Hill (jillian.hill@mrc.ac.za)
Marina de Waal (mdewaal@fedicsunit.co.za)
Lucinda Dallais (luc1_za@yahoo.com)
Zulfa Abrahams (zabrahams@brhc.com)
Carl Lombard (carl.lombard@mrc.ac.za)
Estelle V Lambert (vicki.lambert@uct.ac.za)

Version: 4 Date: 19 August 2015

Author's response to reviews: see over
The Editor,
BMC Public Health

20 August, 2015

Dear Editor,

Re: Did HealthKick, a randomised controlled trial primary school nutrition intervention improve dietary quality of children in low-income settings in South Africa?

Please find the corrected manuscript. We have addressed the comments of the reviewers.

Thank you.

Dr Nelia Steyn
20 August, 2015